Author’s response to reviews

Title: Growth from birth to two years in a cohort of children diagnosed with CF by newborn screening

Authors:
Katie Patterson (katiepatterson.doctors@gmail.com)
Theocharis Kyriacou (t.kyriacou@keele.ac.uk)
Maya Desai (maya.desai@nhs.net)
William Carroll (will.carroll@nhs.net)
Francis Gilchrist (francis.gilchrist@uhnm.nhs.uk)

Version: 4 Date: 13 Sep 2019

Author’s response to reviews:
The following issues have been addressed:

1) As a member of the editorial board (Associate Editor) of this journal, in order to ensure transparency, please declare this in the Competing Interests section of the Declarations.

2) If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations.

3) Please rename "Contributors Statement" to Authors' contributions.